

Author's response to reviews

Title: Antimicrobial susceptibility patterns of Ureaplasma species and Mycoplasma hominis in pregnant women

Authors:

Mathys J Redelinghuys (shanered72@gmail.com)
Marthie M Ehlers (marthie.ehlers@up.ac.za)
Andries W Dreyer (aw.dreyer@gmail.com)
Hennie A Lombaard (hennie.lombaard@up.ac.za)
Marleen M Kock (marleen.kock@up.ac.za)

Version: 3 Date: 17 March 2014

Author's response to reviews: see over
Executive editor
BMC Infectious Diseases

Dear Prof Philippa Harris

SUBMISSION OF FULL LENGTH ARTICLE ENTITLED: “ANTIMICROBIAL SUSCEPTIBILITY PATTERNS OF UREAPLASMA SPECIES AND MYCOPLASMA HOMINIS IN PREGNANT WOMEN”

Thank you for the valuable comments and recommendations on our article. Please find the revised manuscript for possible publication in the Journal: BMC Infectious Diseases. The abovementioned manuscript: “Antimicrobial susceptibility patterns of Ureaplasma species and Mycoplasma hominis in pregnant women” or information has not been published as the focus of a research article before and has not been submitted elsewhere for publication. All the authors and recognised parties have agreed to the submission of this article.

According to the editor’s suggestions, the following have been addressed:

1. More references have been included in the introduction (Par 1) with regards to the pathogenesis of U. urealyticum and U. parvum.
2. The ages of the women are normally distributed. This has been stated, together with the median, in the first paragraph of the results.
3. The general statement on the pricing of azithromycin has been corrected (Discussion, par 7).
4. The spelling of Neisseria gonorrhoeae has been corrected (Discussion, par 9).

The manuscript has also been edited. Please note that the letter from the editor has been attached.

Kind regards
Mr MJ Redelinghuys
PhD student in Medical Microbiology
Corresponding author